Author's response to reviews

Title: Different risks for childhood stunting according to the level of food insecurity in the household: a cross-sectional study in a rural community of Southeastern Kenya

Authors:

Chisa Shinsugi (chisa.shinsugi@gmail.com)
Masaki Matsumura (masaki@nagasaki-u.ac.jp)
Mohamed Karama (mhmdkarama@yahoo.com)
Junichi Tanaka (jtanaka@nagasaki-u.ac.jp)
Mwatasa Changoma (jtasa06@gmail.com)
Satoshi Kaneko (skaneko@nagasaki-u.ac.jp)

Version: 2
Date: 23 September 2014

Author's response to reviews:

Editorial Office, BMC Pediatrics

September 22nd, 2014

Dear Editor:

We are submitting our manuscript entitled “Different risks for childhood stunting according to the level of food insecurity in the household: a cross-sectional study in a rural community of Southeastern Kenya” to your journal.

The contents of this manuscript have not been copyrighted or published previously, are not under consideration for publication elsewhere, and will not be copyrighted, submitted, or published elsewhere while acceptance by the Journal is under consideration.

The results in this study can provide an opportunity for discussion regarding prevention of child stunting in Africa. In this sense, BMC Pediatrics is the most appropriate journal for our study.

We would be grateful if the manuscript could be reviewed and considered for publication in BMC pediatrics.

Sincerely yours,

Satoshi Kaneko, M.D., M.P.H., PhD
Professor,
Department of Ecoepidemiology,
Institute of tropical Medicine, Nagasaki University &
Graduate School of International Health Development, Nagasaki University